Author's response to reviews

Title: An assessment of the health managers and providers response to output-based aid (OBA) voucher program in Kenya: A qualitative study

Authors:

Rebecca Njuki (nrebecca@popcouncil.org)
Timothy Abuya (tabuya@popcouncil.org)
James Kimani (jimani@popcouncil.org)
Lucy Kanya (lkanya@popcouncil.org)
Allan Korongo (allan.korongo@gmail.com)
Piet Bracke (piet.bracke@ugent.be)
Charlotte Warren (cwarren@popcouncil.org)

Version: 2 Date: 13 September 2013

Author's response to reviews: see over
To the Editor
BMC Public Health
13th September 2013

Dear Editor,

RE: Re-submission of a manuscript titled an assessment of the health managers and providers response to output-based aid (OBA) voucher program in Kenya: A qualitative study

We are resubmitting the manuscript titled “An assessment of the health managers and providers response to output-based aid (OBA) voucher program in Kenya: A qualitative study”. We confirm that we have adhered to the RATS guidelines for reporting the qualitative aspect of the study as defined in http://www.biomedcentral.com/authors/rats.

We appreciate your feedback that the manuscript does not fit the scope of BMC Public Health Journal, we request that you transfer this manuscript to the suggested BMC Health Services Research Journal for further consideration.

Yours sincerely,

Rebecca Njuki